Author's response to reviews

Title: The 12-item World Health Organization Disability Assessment Schedule II (WHO-DAS II): A nonparametric item response analysis

Authors:

Juan V Luciano (jvluciano@sjd-ssm.com)
Jose L Ayuso-Mateos (joseluis.ayuso@uam.es)
Jaume Aguado (iaquado@pssjd.org)
Ana Fernandez (afernandez@pssjd.org)
Antoni Serrano-Blanco (aserrano@pssjd.org)
Miquel Roca (mroca@uib.es)
Josep M Haro (jmharo@fsjd.org)

Version: 4 Date: 17 May 2010

Author's response to reviews:

Dear Dr. Graham,

Please, find enclosed a new version of the paper “The 12-item World Health Organization Disability Assessment Schedule II (WHO-DAS II): A nonparametric item response analysis” (MS: 1838976187346862), by Juan V. Luciano, José L. Ayuso-Mateos, Jaume Aguado, Ana Fernández, Antoni Serrano-Blanco, Miquel Roca, and Josep M. Haro.

We have made some minor changes in the manuscript following your recommendations. However, we find impossible to increase the quality of Figures 2 and 4, as you required ("Figures: at 100% the text of Figures 2 and 4 is not readable. We recommend you upload a higher resolution version of the figures").

The software we employed to make the statistical analyses do not let us to obtain Figures with greater quality. Please, could you modify the figures for us?

Thanks in advance. Yours sincerely,

Juan V. Luciano